Author’s response to reviews

Title: EFFECT OF TWO ALTERNATIVE METHODS OF POOLING SPUTUM PRIOR TO TESTING WITH XPERT MTB/RIF

Authors:

TBPhuong Nguyen (phuong.nguyen@sydney.edu.au)
Thu Anh Nguyen (thuanh.nguyen@sydney.edu.au)
Van Son Nguyen (drson.camau@gmail.com)
Vitali Sintchenko (vitali.sintchenko@health.nsw.gov.au)
Jennifer Ho (jho@gmp.usyd.edu.au)
Greg J. Fox (gregory.fox@sydney.edu.au)
Viet Nhung Nguyen (vietnhung@yahoo.com)
Guy B. Marks (g.marks@unsw.edu.au)

Version: 4 Date: 21 Jan 2019

Author’s response to reviews:

21st January, 2019

Oliver Cocks
Editorial Manager
BMC Infectious Diseases

Dear Mr Cocks,

Re: INFD-D-18-01325

EFFECT OF TWO ALTERNATIVE METHODS OF POOLING SPUTUM PRIOR TO TESTING WITH XPERT MTB/RIF
Revision 4

Thank you for provisionally accepting this for publication. We have addressed the issues raised in your recent correspondence (see below) and, as requested, uploaded only a clean copy, without track changes.

Editor comments

1. There is a typographical error in your “Abstract” section heading on line 40. Please revise.

Additionally, please remove the section heading “Objective” from your abstract. Only the subsections - Background, Methods, Results, Conclusions – should be present in the Abstract.

Done

2. As indicated in our submission guidelines, please include/revise the section headings of your main text to the following: Background, Methods, Results, Discussion, and Conclusion.

Done

3. Please relocate Table 1 so that it is after the References section.

Done

4. Please include all of the following subsections in your Declarations:

   a. Ethics approval and consent to participate
   b. Consent for publication
c. Availability of data and materials

d. Competing interests

e. Funding

f. Authors’ contributions

g. Acknowledgements

Done. Information on “Consent for publication” has been added.

5. Please include the initials and contributions of all authors individually in the Authors’ contributions subsection. Forgoing the listing of individual authors in favour of stating “all authors” contributing to one or more aspects is not accepted.

Done

6. In your Competing interests subsection please declare that author Greg Fox is an Editorial Board Member if BMC Infectious Diseases.

Done

7. Please remove all mention of funding from the Acknowledgments subsection. Information relating to funding should be mentioned solely in the Funding subsection.

Done

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done
Yours sincerely,

Nguyen Thi Bich Phuong